Common E-L Commands

Artifact references copy-as-kill C-z x With cursor on reference or artifacts / drafts plex will copy references. C-y to insert.

follow-link C-z^

Artifacts edit-artifact C-z C-e With cursor on reference, or in artifacts plex, move cursor to correct screen before confirming. Creates new draft examine-artifact C-z C-x Read only pop-to-artifact Buffer on artifact in another window derive-from Force derivation delete-one-artifact C-z C-d If not referenced, no active draft delete-artifact-and-references C-z M-d delete-artifacts-in-region C-z M-D In window on artifacts plex archive Stores down. If 2 or 3 C-us stores up as well import-archive 1 or 3 C-us to override timestamp. 2 or 3 C-us to override creator export-derivative Copy derivative to Unix file

Drafts commit-one-draft C-u C-z C-s Just this draft and don’t schedule anything commit-one-draft C-z C-s Just this draft commit-draft-and-references C-z M-s This and down commit-draft-and-references-and-referencers This, up and down C-z M-C-s create-draft C-z C-c Creates draft and buffer for it use C-x C-b name to get to buffer create-draft-and-reference C-z M-c like create-draft but also inserts reference at point in buffer delete-draft Also C-x k on only buffer on draft abandon-draft C-z C-a To hand off to another user takeover-draft C-z e To take over abandoned draft revert-draft Restore buffer to last autosave find-successor-draft In drafts plex; Of some artifact

Notices and Pending display-notice C-z C-n Display text of notice delete-notice C-z M-n Delete selected notice act-on-pending-item C-z C-p

LA_{TEX}

latex-log Examine LATEXlog preview Preview with xdvi ps-file Create Postscript file hardcopy Print document or latex-piece section

Unix = C highlight-region C-z C-h Affects code going into .h file execute A unix-program dbx A unix-program stop-at-c-line C-z C-@

dbx breakpoint at current line of artifact

examine-problems Like display-notice but for artifact pop-to-artifact-per-cursor C-z ‘ While on display-notice/problems line pop up source artifact and highlight

Search, Diff and Merge re-search-artifacts-recursively C-z s Search artifact and descendants diff-buffers unbutton C-z u Removes highlighting, C-u C-z u removes highlighted text diff-artifacts diff-selected-artifact diff-artifacts-recursively merge-artifacts Also affects references to the artifacts merge-selected-artifact merge-artifact-into-draft merge-drafts

Plexes view-plex C-z C-v +1 C-u for special filter, +2 C-us for special format C-] for no more filter or format selections, M-] to restart selections with changes, M-C-] to restart from beginning state refilter C-z C-i reformat C-z C-o

System management newuser Currently automatic delete-user stop-job-server start-job-server After stop-job-server please goodbye-hostport Pings port to check if alive, removes if not find-missing-draft After crash, then goodbye-hostport on returned port, will release draft owned by crashed job to your use